Author's response to reviews

Title: Association between CYP19 gene SNP rs2414096 Polymorphism and polycystic ovary syndrome in Chinese women

Authors:

Jia-Li Jin (jjllily@126.com)
Jing Sun (sjinju@163.com)
Yun-xia Cao (caoyunxia6@126.com)
Xiao&#8211;Ke Wu (xiaokewu@nju.edu.cn)
Feng-jing Liang (qgq149@sina.com)
Hai-xiang Sun (stevensun@sohu.com)
Lu Ke (kbb9832@163.com)
Long Yi (yilong@nju.edu.cn)
Yong Wang (yongwang@nju.edu.cn)

Version: 2 Date: 7 August 2009

Author's response to reviews:

Dear Editor,

I would like to submit my MS (ID1512193024296179) to BMC Medical Genetics. If you have any questions, Please do not hesitate to contact me.

Regards,
Yong WANG, Ph.D.